Author’s response to reviews

Title: Insulin autoimmune syndrome induced by exogenous insulin injection: a four-case series

Authors:

Yimin Shen (shen06@zju.edu.cn)
Xiaoxiao Song (xsong@163.com)
Ren YueZhong (renyuez@zju.edu.cn)

Version: 2 Date: 15 Dec 2019

Author’s response to reviews:

Rebuttal letter

Dear editor,

Thank you for your review of and response to our manuscript “Insulin autoimmune syndrome induced by exogenous insulin injection: A four-case series”. We are enclosing a revised manuscript in which we have addressed the comments of the reviewers, and we hope that this manuscript will meet the standards for publication. The authors thank the editors and the reviewers for their pertinent and constructive comments.

Editor Comments:

1. Please move the keywords section to after the abstract and before the background.

Answer: Thanks for your suggestion. We have moved them in the revised version.

2. Please ensure the formatting of your manuscript follows the sections provided in our submission guidelines (https://bmcendocrdisord.biomedcentral.com/submission-guidelines/preparing-your-manuscript/case-report)

Answer: Thanks for your advice. We ensure the formatting of manuscript follows the sections provided in submission guidelines.

3. In the consent for publication section, please clarify if written informed consent for publication was obtained from all patients.
Answer: Thanks for your reminds. We have clarified in publication section.

4. Tables 1-3 are duplicated in the manuscript. Please amend this.

Answer: Thanks for your reminds. We have deleted them in the manuscript.

5. Please remove any additional files you do not want published alongside your manuscript, including cover/response letters.

Answer: Thanks for your suggestion. We have removed cover and response letters.

6. We note that the current submission contains some textual overlap with other previously published works, in particular:

Page 2 lines 30-41:


While we understand that you may wish to express some of the same ideas contained in these publications, please be aware that we cannot condone the use of text from previously published work.

If there is overlap in the Methods section, please ensure to summarize the methods and cite the source.

Please re-phrase these sections to minimise overlap.

Answer: Thanks for your reminds. We have re-phrased the sentence in the revised version.